Author's response to reviews

Title: A process evaluation of a "physical activity pathway" in the primary care setting

Authors:

   Fiona C Bull (Fiona.Bull@uwa.edu.au)
   Karen E Milton (K.Milton@lboro.ac.uk)

Version: 7 Date: 6 August 2010

Author's response to reviews:

Dear editor

We have copied and pasted the title and abstract directly from the manuscript into the submission system. Any discrepancies may be due to the text format being modified during upload. If there continues to be a problem, could you please use the title and abstract as they appear in the manuscript.

Many thanks

Karen Milton